Author's response to reviews

Title: Conservation of resources theory and research use in health systems

Authors:

Celeste Alvaro (Celeste.Alvaro@dal.ca)
Renée F Lyons (RLyons@bridgepointhealth.ca)
Grace Warner (Grace.Warner@dal.ca)
Stevan E Hobfoll (Stevan.Hobfoll@rush.edu)
Patricia J Martens (Pat_Martens@cpe.umanitoba.ca)
Ronald Labonté (rlabonte@uottawa.ca)
E. Richard Brown (erbrown@ucla.edu)

Version: 6 Date: 4 October 2010

Author's response to reviews:

Please find attached the requested revisions for October 4, 2010.